Author's response to reviews

Title: A study of the number of physicians registered in Jamaica and their countries of training.

Authors:

Jennifer M Knight-Madden (jennifer_knight@yahoo.com)
Robert H Gray (rhgray@cwjamaica.com)

Version: 5 Date: 28 November 2008

Author's response to reviews:

We made very minor modifications, such as removing extra spaces between words and correcting typographical errors. There were no substantive changes made. For example